
\documentclass[12pt]{article}

%%%%%%%%% Style taken from the Electronic Journal of Combinatorics: http://www.combinatorics.org/files/e-jc.sty

\setlength{\textwidth}{6.3in}
\setlength{\textheight}{8.7in}
\setlength{\topmargin}{0pt}
\setlength{\headsep}{0pt}
\setlength{\headheight}{0pt}
\setlength{\oddsidemargin}{0pt}
\setlength{\evensidemargin}{0pt}

\usepackage[utf8]{inputenc}

\usepackage{amssymb}
\renewcommand{\geq}{\geqslant}
\renewcommand{\leq}{\leqslant}
\renewcommand{\ge}{\geqslant}
\renewcommand{\le}{\leqslant}


\newenvironment{corrigendum}[1]%
{\clearpage\newpage\section*{Corrigendum -- submitted #1}}{}

\def\reusenumber#1#2{\setcounter{#1}{#2}\addtocounter{#1}{-1}}

\def\specs#1#2#3{\def\papno{#1}\def\volno{#2}\def\volyear{#3}}
\ifx\volno\undefined\def\volno{}\fi
\ifx\volyear\undefined\def\volyear{}\fi
\ifx\papno\undefined\def\papno{}\fi
\def\dateline#1#2{{\small Submitted: #1; Accepted: #2; Published: XX}}
\def\dateline#1#2#3{{\small Submitted: #1; Accepted: #2; Published: #3}}

\newfont{\footsc}{cmcsc10 at 8truept}
\newfont{\footbf}{cmbx10 at 8truept}
\newfont{\footrm}{cmr10 at 10truept}
%\renewcommand{\ps@plain}{%
%\renewcommand{\@oddfoot}{\footsc the electronic journal of combinatorics
%  {\footbf\volno} (\volyear), \#\papno\hfil\footrm\thepage}}
\pagestyle{plain}

%%%  Squashing the bibliography together a bit
  \newlength{\BiblioSpacing}
  \setlength{\BiblioSpacing}{0.4ex plus 0.1ex minus 0.1ex}
  \newcommand{\SquashBibFurther}{%
     \setlength{\BiblioSpacing}{0.2ex plus 0.1ex}}

  \let\oldthebibliography=\thebibliography
  \let\endoldthebibliography=\endthebibliography
  \renewenvironment{thebibliography}[1]{%
    \begin{oldthebibliography}{#1}%
      \setlength{\parskip}{\BiblioSpacing}
      \setlength{\itemsep}{\BiblioSpacing}
  }%
  {%
    \end{oldthebibliography}%
  }

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage{amsthm,amsmath,amssymb}
\usepackage{graphicx}
\usepackage{cite}
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}

\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem*{definition}{Definition}
\newtheorem{fact}{Fact}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\R{\mathbb{R}}
\def\Z{\mathbb{Z}}
\def\C{\mathbb{C}}
\def\N{\mathbb{N}}
\def\F{\mathbb{F}}
\def\Q{\mathbb{Q}}
\def\infinity{  \infty }
\def\noteq{\neq}
\def\parts{\operatorname{par}}
\def\summ{\operatorname{sum}}
\def\len{\operatorname{len}}
\def\fst{\operatorname{fst}}
\def\d{\mbox{-}}

\title{\bf Counting words with Laguerre series}

\author{Jair Taylor\\
\small Department of Mathematics\\[-0.8ex]
\small University of Washington\\[-0.8ex] 
\small Seattle, WA, U.S.A.\\
\small\tt jptaylor@uw.edu\\
}

\date{\dateline{June 26, 2013}{March 11, 2014}\\
\small Mathematics Subject Classifications: 05A05, 05A15, 33C45}

\begin{document}
\begin{thebibliography}{10}

\bibitem{Steingrimsson}
Eric Babson and Einar Steingr{\'{\i}}msson.
\newblock Generalized permutation patterns and a classification of the
  {M}ahonian statistics.
\newblock {\em S\'em. Lothar. Combin.}, 44:Art. B44b, 18 pp. (electronic),
  2000.

\bibitem{Bernini}
Antonio Bernini, Luca Ferrari, and Renzo Pinzani.
\newblock Enumeration of some classes of words avoiding two generalized
  patterns of length three.
\newblock {\em J. Autom. Lang. Comb.}, 14(2):129--147, 2009.

\bibitem{Mississippi}
Gunnar Blom, Jan-Eric Englund, and Dennis Sandell.
\newblock The {M}ississippi problem.
\newblock {\em American Statistician}, 52(1):49--50, 1998.

\bibitem{Claesson}
Mireille Bousquet-M{\'e}lou, Anders Claesson, Mark Dukes, and Sergey Kitaev.
\newblock Unlabeled {$(2+2)$}-free posets, ascent sequences and pattern
  avoiding permutations.
\newblock In {\em 21st {I}nternational {C}onference on {F}ormal {P}ower
  {S}eries and {A}lgebraic {C}ombinatorics ({FPSAC} 2009)}, Discrete Math.
  Theor. Comput. Sci. Proc., AK, pages 216--228. Assoc. Discrete Math. Theor.
  Comput. Sci., Nancy, 2009.

\bibitem{Burstein}
Alexander Burstein.
\newblock {\em Enumeration of words with forbidden patterns}.
\newblock ProQuest LLC, Ann Arbor, MI, 1998.
\newblock Thesis (Ph.D.)--University of Pennsylvania.

\bibitem{Mansour5}
Alexander Burstein and Toufik Mansour.
\newblock Counting occurrences of some subword patterns.
\newblock {\em Discrete Math. Theor. Comput. Sci.}, 6(1):1--11, 2003.

\bibitem{Mansour2}
Alexander Burstein and Toufik Mansour.
\newblock Words restricted by 3-letter generalized multipermutation patterns.
\newblock {\em Ann. Comb.}, 7(1):1--14, 2003.

\bibitem{Wilf}
Alexander Burstein and Herbert~S. Wilf.
\newblock On cyclic strings without long constant blocks.
\newblock {\em Fibonacci Quart.}, 35(3):240--247, 1997.

\bibitem{Carlitz}
L.~Carlitz.
\newblock Restricted compositions.
\newblock {\em Fibonacci Quart.}, 14(3):254--264, 1976.

\bibitem{CSV}
L.~Carlitz, Richard Scoville, and Theresa Vaughan.
\newblock Enumeration of pairs of sequences by rises, falls and levels.
\newblock {\em Manuscripta Math.}, 19(3):211--243, 1976.

\bibitem{Doron2}
Anne~E. Edlin and Doron Zeilberger.
\newblock The {G}oulden-{J}ackson cluster method for cyclic words.
\newblock {\em Adv. in Appl. Math.}, 25(2):228--232, 2000.

\bibitem{Even}
S.~Even and J.~Gillis.
\newblock Derangements and {L}aguerre polynomials.
\newblock {\em Math. Proc. Cambridge Philos. Soc.}, 79(1):135--143, 1976.

\bibitem{Doron3}
Dominique Foata and Doron Zeilberger.
\newblock Laguerre polynomials, weighted derangements, and positivity.
\newblock {\em SIAM J. Discrete Math.}, 1(4):425--433, 1988.

\bibitem{Gessel}
Ira~M. Gessel.
\newblock Generalized rook polynomials and orthogonal polynomials.
\newblock In D.~Stanton, editor, {\em q-Series and Partitions}, pages 159--176.
  Springer-Verlag, New York, 1989.

\bibitem{Goulden-Jackson1}
I.~P. Goulden and D.~M. Jackson.
\newblock An inversion theorem for cluster decompositions of sequences with
  distinguished subsequences.
\newblock {\em J. London Math. Soc. (2)}, 20(3):567--576, 1979.

\bibitem{Goulden-Jackson2}
Ian~P. Goulden and David~M. Jackson.
\newblock {\em Combinatorial enumeration}.
\newblock Dover Publications, Inc., Mineola, NY, 2004.
\newblock With a foreword by Gian-Carlo Rota, Reprint of the 1983 original.

\bibitem{Mansour1}
Silvia Heubach and Toufik Mansour.
\newblock {\em Combinatorics of compositions and words}.
\newblock Discrete Mathematics and its Applications (Boca Raton). CRC Press,
  Boca Raton, FL, 2010.

\bibitem{Knopfmacher}
Arnold Knopfmacher and Helmut Prodinger.
\newblock On {C}arlitz compositions.
\newblock {\em European J. Combin.}, 19(5):579--589, 1998.

\bibitem{MacMahon}
Percy~A. MacMahon.
\newblock {\em Combinatory analysis}.
\newblock Two volumes (bound as one). Chelsea Publishing Co., New York, 1960.

\bibitem{ortho}
Wilhelm Magnus, Fritz Oberhettinger, and Raj~Pal Soni.
\newblock {\em Formulas and theorems for the special functions of mathematical
  physics}.
\newblock Third enlarged edition. Die Grundlehren der mathematischen
  Wissenschaften, Band 52. Springer-Verlag New York, Inc., New York, 1966.

\bibitem{Mansour6}
Toufik Mansour.
\newblock Restricted 132-avoiding {$k$}-ary words, {C}hebyshev polynomials, and
  continued fractions.
\newblock {\em Adv. in Appl. Math.}, 36(2):175--193, 2006.

\bibitem{Doron1}
John Noonan and Doron Zeilberger.
\newblock The {G}oulden-{J}ackson cluster method: extensions, applications and
  implementations.
\newblock {\em J. Differ. Equations Appl.}, 5(4-5):355--377, 1999.

\bibitem{Pollard}
Harry Pollard.
\newblock Representation of an analytic function by a {L}aguerre series.
\newblock {\em Ann. of Math. (2)}, 48:358--365, 1947.

\bibitem{Smirnov}
N.~V. Smirnov, O.~V. Sarmanov, and V.~K. Zaharov.
\newblock A local limit theorem for the number of transitions in a {M}arkov
  chain and its applications.
\newblock {\em Dokl. Akad. Nauk SSSR}, 167:1238--1241, 1966.

\bibitem{szasz}
Otto Sz{\'a}sz and Nelson Yeardley.
\newblock The representation of an analytic function by general {L}aguerre
  series.
\newblock {\em Pacific J. Math.}, 8:621--633, 1958.

\bibitem{weniger}
Ernst~Joachim Weniger.
\newblock On the analyticity of {L}aguerre series.
\newblock {\em J. Phys. A}, 41(42):425207, 43, 2008.

\bibitem{zeng}
Jiang Zeng.
\newblock Weighted derangements and the linearization coefficients of
  orthogonal {S}heffer polynomials.
\newblock {\em Proc. London Math. Soc. (3)}, 65(1):1--22, 1992.

\end{thebibliography}


\maketitle
\begin{abstract}
We develop a method for counting words subject to various restrictions by finding a combinatorial interpretation for a product of weighted sums of Laguerre polynomials with parameter $\alpha = -1$.  We describe how such a series can be computed by finding an appropriate ordinary generating function and applying a certain transformation. We use this technique to find the generating function for the number of $k$-ary words avoiding any vincular pattern that has only ones, as well as words cyclically avoiding vincular patterns with only ones whose runs of ones between dashes are all of equal length.

\bigskip\noindent \textbf{Keywords:} Laguerre polynomial, orthogonal polynomial, combinatorics on words, pattern avoidance
\end{abstract}
\maketitle\section{Introduction}

Consider the following simple problem.  How many arrangements of the word ``WALLAWALLA'' are there with no LLL, AAA or WW as consecutive subwords?  Perhaps surprisingly, the answer can be calculated by performing a certain integral.  In fact, there are \[\int_0^\infty e^{-t}\left(\frac{1}{24}t^4 - t^2 + t\right)^2  \left(\frac{1}{2}t^2 - t\right) \, dt = 1584\] such words, as we will see.  More generally, we will develop a technique based on a result of Gessel\cite{Gessel} by which the generating function for the number of words subject to various conditions can be found by integrating a product against $e^{-t}$, $t\geq 0$.  Since $\int_0^\infty e^{-t}t^n \, dt = n!$, this amounts to applying the linear functional $\Phi: \R[t] \mapsto \R$ given by $t^n \mapsto n!$.

Define a {\it word} $W$ on an alphabet $S$ to be an ordered list $s_1 \cdots s_n$ of letters $s_i \in S$.  If the alphabet $S$ is the set $\N$ of positive integers, then $W = s_1 \cdots s_n$ is sometimes called a \emph{composition} of $s_1 + \cdots + s_n$.  A \emph{subword} of $W$ is a word $s_l s_{l+1} \cdots s_{l+m}$.   Note that we require the indices in a subword to be consecutive, while some authors do not.  Define a {\itshape factorization} $\phi$ on $S$ to be an ordered list $(\phi_1)(\phi_2) \ldots (\phi_k)$ of non-empty words $\phi_1, \ldots, \phi_k$ using letters from $S$.  We call $\phi_1, \ldots, \phi_k$ the {\it factors} or {\it parts} of $\phi$.  We abuse notation slightly by identifying a word $W$ with the factorization $(W)$ in one part, and the empty word with the factorization having no parts, writing $\emptyset$ for both.  

Given a factorization $\phi$ on $S$ and $T \subseteq S$, let $\phi|_T$ be the factorization whose parts are the maximal subwords of the parts of $\phi$ that use only letters from $T$, the parts of $\phi|_T$ being ordered by their occurrence in $\phi$.  We call $\phi|_T$ the restriction of $\phi$ to $T$.  For example, if $S = \{a,b\}$ and $T = \{a\}$, then the restriction of the factorization $(aabba)(aab)(b)(aaa)$ to $T$ is $\phi|_T = (aa)(a)(aa)(aaa)$, while $\phi|_{\{b\}} = (bb)(b)(b)$.  If $\phi$ contains no letters from $T$, we define $\phi|_T$ to be the empty factorization.

The methods given here will allow us to count words where only certain restrictions can occur.  For example, consider the words $W$ so that for each $s \in S$, the factors of $W|_{\{s\}}$ are all length one. This is equivalent to requiring that $W$ have no adjacent, identical letters.  Such words are sometimes called {\it Carlitz}, after Leonard Carlitz.  Thus ``PEAR'' is Carlitz, but ``APPLE'' is not.

More generally, let $A_1$ and $A_2$ be two sets of factorization on disjoint alphabets $S_1$, $S_2$, respectively.  Let $S = S_1 \cup S_2$, and denote by $A_1* A_2$ the set of factorizations $\phi$ of words on $S$ so that $\phi|_{S_1} \in A_1$ and $\phi|_{S_2} \in A_2$.  For example, if $A_1 = \{\emptyset, (a),(a)(a),(a)(a)(a), \ldots\}$, $B_2 = \{\emptyset, (b),(b)(b),(b)(b)(b), \ldots\}$, then $A_1 * A_2$ is the set of factorizations on $\{a,b\}$ so that each factor is a Carlitz word.  Thus $(ab)(a)(a)(baba) \in A_1 * A_2$, but $(b)(aab)(ba) \notin A_1 *A_2$.

Perhaps surprisingly, the problem of counting words in $A_1*A_2$ can be reduced to multiplication of certain power series.  If $A$ is a set of factorizations, we define a weight $w$ on $A$ to be a polynomial-valued function on $A$ that obeys a certain combinatorial restriction.  Then we will define a power series $f_{A,w}(t)$, the associated {\it Laguerre series} for $A$, in terms of the generalized Laguerre polynomials with parameter $\alpha = -1$.  The key fact we will use is the rule (Theorem~\ref{key}) \[f_{A_1*A_2,w}(t) = f_{A_1,w}(t) \cdot f_{A_2,w}(t).\]  Furthermore, we will show in Proposition~\ref{phifact} that $\Phi(f_{A,w}(t))$ gives the weight of all factorizations in $A$ with one or no parts, which we identify with words. By applying $\Phi$ to a product of Laguerre series we may count a variety of sets of restricted words, especially when the restrictions are on the length of runs of particular letters.

In Section~3, we describe the transformation $T$ that turns certain ordinary generating functions into Laguerre series.  The transformation can be described in terms of the Laplace transform, and so can be easily implemented in mathematics software packages.  We use $T$ to determine the Laguerre series for different sets of factorizations $A$, and use them to derive formulas and generating functions which count words that obey various restrictions.

In particular, we can use this technique to analyze certain pattern avoidance problems. A vincular, or generalized, pattern is a pattern with dashes such as $13 \d 2$.  This is a generalization of classical permutation patterns where the dashes are used to indicate that the numbers on either side are not required to be adjacent, but all others are.  These patterns were first studied by Babson and Steingr\'{i}msson \cite{Steingrimsson}, who showed that many statistics of interest can be classified in terms of vincular patterns.  The term {\itshape vincular} itself was coined by Claesson in \cite{Claesson}, from the Latin {\it vinculare}, to bind.  Words avoiding vincular patterns are studied in \cite{Bernini,Burstein, Mansour2,Mansour1,Mansour5, Mansour6}.  In this paper we will study vincular patterns with all ones, such as $111\d 11$.  Although such patterns are useless in the context of permutations, where only the pattern $1$ can be contained, they are meaningful in the context of general words on the alphabet $\N$ where letters may be repeated.  Since all of our patterns will have only ones, we give the definition in this context.  We say that a word $W = s_1 \cdots s_l$, with each $s_i$ in some alphabet $S$, contains a vincular pattern $\tau = 1^{m_1}\d\cdots\d 1^{m_n}$ if there is a subsequence of $W$ consisting of $m = m_1 + \ldots + m_n$ identical letters of which the first $m_1$ are consecutive, the next $m_2$ are consecutive, and so on.  For example, a word contains $111\d 11$ if it has five appearances of the same letter in the word so that the first three and the last two are consecutive.  Thus the word $W = aaaabbaab$ contains $111\d 11$ but $aabbaaaa$ does not.  Formally, we say that $W$ contains $\tau$ if there are indices $1 \leq i_1 < i_2 < \ldots < i_m \leq l$ with $s_{i_1} = \ldots = s_{i_m}$ and $i_{j+1} - i_j = 1$ for $j \neq m_1, m_1 + m_2, \ldots, m_1 + \ldots + m_{n-1}$.  Otherwise, we say that $W$ avoids $\tau$.    

In Section~$4$, we give a formula to calculate the generating function for the number of words avoiding any such vincular pattern with only ones.  This formula involves the use of the maps $T$ and $\Phi$, but these can be easily calculated.  For example, we can use Sage to compute the the generating function $\sum_W x^{\len(W)}$ where the sum is taken over all ternary words $W$ avoiding the pattern $11  \d 11$, where $\len(W)$ is the ${\it length}$ of $W$, the number of letters counting multiplicity:

\[\frac{6 \, x^{7} - 6 \, x^{6} + 6 \, x^{5} - 2 \, x^{4} - 5 \, x^{3} + 9 \, x^{2} - 5 \, x + 1}{16 \, x^{4} - 32 \, x^{3} + 24 \, x^{2} - 8 \, x + 1} = 1 + 3 x + 9 x^{2} + 27 x^{3} + 78 x^{4} + 222 x^{5}+ \ldots.\]

We  give a cyclic version of this result in Section~$5$ for the case of patterns $1^m \d 1^m \d \cdots \d 1^m$, where all runs of ones are the same length.  This gives the generating functions for words so that any cyclic permutation of their letters avoids such a pattern.  This generalizes a result of Burstein and Wilf \cite{Wilf} which gives the generating function for the number of words cyclically avoiding $1^m$.
%%%
\section*{Acknowledgments}
The author would like to thank Sara Billey, Ira Gessel, Silvia Heubach, Yannick Van Huele, Toufik Mansour, Brendan Pawloski, Austin Roberts, Byron Schmuland and William Stein for their insights and suggestions, as well as the anonymous reviewer for helpful comments including the simplified proof of Lemma 2.2 that appears here. We also thank the reviewers of an extended abstract version of this paper which appeared in the proceedings of Formal Power Series and Algebraic Combinatorics (FPSAC) conference in 2013.
%%%
\section{Laguerre polynomials and Laguerre series} 
Define the polynomials $l_k(t)$ by their generating function 
\begin{align}\label{laguerregf}
\sum_{k=0}^\infty l_k(t) x^k = e^{\frac{tx}{1+x}}.
\end{align} The first few such polynomials are
\begin{align*}
l_0(t)  &= 1\\
l_1(t) &= t\\
l_2(t) &= \frac{1}{2}t^2 - t\\
l_3(t) &= \frac{1}{6}t^3 - t^2 + t.
\end{align*}
These polynomials are a form of Laguerre polynomial.  Specifically, $l_k(t) = (-1)^k L_k^{(-1)}(t)$ where $$L_k^{(\alpha)}(t) = \sum_{i=0}^k (-1)^i \binom{k + \alpha }{k - i} \frac{t^i}{i!}$$ defines the generalized Laguerre polynomials. They have been found to have a number of interesting combinatorial properties, beginning with the use of the classical Laguerre polyomials $L_k^{(0)}$ by Even and Gillis~\cite{Even} to count generalized derangements.  This was later extended by Foata and Zeilberger who use $\alpha$ to keep track of the number of cycles \cite{Doron3}.  For our purpose, we will take $\alpha = -1$.

Recall that a word on $[k] = \{1,2, \ldots, k\}$ is called $k$-ary, and a word in which no two adjacent letters are the same is called a Carlitz word.  Ira M. Gessel \cite[Section 6]{Gessel} used his theory of generalized rook polynomials to find an explicit formula for the number of Carlitz arrangements of a given multiset of letters.  We present here an unlabeled version.

\begin{theorem}\label{fundamental}
Let $\Phi$ be the linear functional on polynomials in $t$ given by $\Phi(t^n) = n!$.  Given nonnegative integers $n_1, \ldots, n_k$, the number of $k$-ary Carlitz words with the letter $i$ used exactly $n_i$ times is \[\Phi\biggl(\prod_{i=1}^k l_{n_i}(t)\biggr).\]
\end{theorem}

For example, in \cite{Mississippi} the authors consider the ``Mississippi Problem''.  How many arrangements of the letters in the word ``MISSISSIPPI'' have no adjacent letters the same?  We can use the preceding theorem to calculate this directly.  There is one $M$, four $I$'s, four $S$'s, and two $P$'s.  So the solution is \[\Phi(l_1(t)l_4(t)l_4(t)l_2(t)) = \int_0^\infty e^{-t} \, \Big( t \Big) \, \left( \frac{1}{24}t^4 - \frac{1}{2}t^3 + \frac{3}{2} t^2 - t\right)^2\left(\frac{1}{2}t^2 - t\right)\, dt = 2016.\]

Our proof of Theorem~\ref{fundamental} does not involve rook polynomials, but is based on the following surprisingly simple expression for the generating function for the number of Carlitz words with a given set of letters.

\begin{lemma}\label{carlitzgf}
Let $\mathcal{C}$ be the set of Carlitz words over the alphabet $\N$.  For a word $W = s_1\cdots s_n$ define its weight to be $w(W) = x_{s_1}\cdots x_{s_n}$, so that the exponent of $x_i$ records the number of occurrences of the letter $i$.  Then 
\begin{align*}\label{smirnov}
\sum_{W \in \mathcal{C}} w(W) = \frac{1}{1-\sum_i \frac{x_i}{1+x_i}}.
\end{align*}
\end{lemma}
Lemma ~\ref{carlitzgf} can be found in the book by Goulden and Jackson \cite[2.4.16]{Goulden-Jackson2}. They attribute it to MacMahon's book from 1915 \cite{MacMahon} and Smirnov, Sarmanov, and Zaharov in (1966) \cite{Smirnov}, and so counting Carlitz words is sometimes known as the Smirnov problem.  The case $x_i = x^i$ was found, apparently independently, by Carlitz in 1977 \cite{Carlitz}.  Asymptotics of Carlitz compositions are investigated using this formula by Knopfmacher and Prodinger in \cite{Knopfmacher}. This result can be generalized to the Carlitz-Scoville-Vaughan Theorem \cite{CSV}, a reciprocity result relating the words for which only a given set of pairs of letters are adjacent to the words where none of these pairs are adjacent. 
\begin{proof}[Proof of Lemma~\ref{carlitzgf}]
Given any word $W$, we can reduce it to a Carlitz word $\hat{W}$ by replacing each maximal string of consecutive identical letters $k$ with a single letter $k$.  For example, if $$W = {12111321123322  },$$ we get the Carlitz word $$\hat{W} = {121321232 }.$$  Furthermore, we can recover $W$ from $\hat{W}$ if we know the length of each substring of identical letters and what order the substrings occur in.  Thus given the set of Carlitz words, we can recover all words by replacing each letter of each Carlitz word by a string of any number of copies of that letter.  

Let $$F(x_1, x_2, \ldots) = \sum_{W \in \mathcal{C}} w(W).$$  Then replacing each $x_i$ by $t_i + t_i^2 + \cdots = t_i/(1-t_i)$ in $F$ gives the weight of all words attained by choosing a number of repetitions for each letter of a Carlitz word over $\N$.  By the discussion of the previous paragraph, this gives the weight of all words over $\N$, and so we have
%%%
\begin{align*}
F\left(\frac{t_i}{1-t_i}, \frac{t_i}{1-t_i}, \ldots\right) = \frac{1}{1 - \sum_{i=1}^\infty t_i}.
\end{align*}
%%%
Setting $t_i = x_i/(1+x_i)$ gives $x_i = t_i/(1-t_i)$, finishing the proof.
\end{proof}
\begin{proof}[Proof of Theorem~\ref{fundamental}]
First note that it is possible to extend the linear functional $\Phi$ to some power series. Specifically, if $f \in \R[t,x_1,x_2,\ldots]$ is a power series with the property that the coefficient of each monomial $x_1^{n_1}\cdots x_k^{n_k}$ is a polynomial in $t$ then define $\Phi(f)$ by letting $\Phi$ fix the variables $x_1, x_2, \ldots$ and act only on $t$. In this case we have \[\Phi(e^{tf}) = \frac{1}{1-f}.\] Then we see
\begin{align*}
\Phi\left(\prod_{i=1}^\infty \exp\left(\frac{tx_i}{1+x_i}\right)\right) &= \Phi\left( \exp\left(t\sum_{i=1}^\infty\frac{x_i}{1+x_i}\right)\right)\\
&= \frac{1}{1 - \sum_{i=1}^\infty \frac{x_i}{1+x_i}}\\
\end{align*}
and if we take the coefficient of $x_1^{k_1} x_2^{k_2} \cdots x_m^{k_m}$ on both sides we get the desired formula by Lemma ~\ref{carlitzgf} and the generating function \eqref{laguerregf}.
\end{proof}

Using Theorem~\ref{fundamental}, it is easy to see combinatorially that 

\begin{align}\label{orthogonality}
  \Phi\left(l_i(t) l_j(t)\right)  = \left\{
     \begin{array}{ll}
       2 & \text{if $i = j$} \\
       1 & \text{if $|i-j| = 1$}\\
		 0 & \text{if $|i-j| > 1$}\\
     \end{array}
   \right.
\end{align} 

\noindent and so the polynomials $l_k(t)$ are ``almost'' orthogonal with respect to $\Phi$.  

Note that $l_k(t)$ is a polynomial of degree $k$; so the matrix of $l_k$'s expanded into powers of $t$ is triangular with no zeroes on the diagonal, and so $\{l_k\}_k$ forms a basis of $\R[t]$.  It is natural to ask, then, what is the expansion of $l_i(t) l_j(t)$ in this basis?  These are known as {\it linearization coefficients}.  The linearization coefficients of general Laguerre polynomials, with $\alpha$ an indeterminate, are known \cite{ Doron3,zeng}, but we will need a combinatorial interpretation of the case $\alpha = -1$.

Denote by $n_{i,j,k}$ the number of factorizations over the alphabet $\{a,b\}$ with $k$ parts and exactly $i$ $a$'s and $j$ $b$'s so that each part is Carlitz.  For example, $n_{2,5,3} = 6$: the possibilities are $(bab)(bab)(b)$, $(babab)(b)(b)$ and the different permutations of these sets of factors.  

\begin{lemma}\label{linearization}
We have, for all $i,j \in \N$, \[l_i(t)l_j(t) = \sum_k n_{i,j,k}l_k(t).\]
\end{lemma}
\begin{proof}
Note that if $p(t) = a_0 + a_1t + \ldots +a_nt^n$ is a polynomial and $\Phi(t^m p(t)) = 0$ for all $m$, then \[a_0m! +a_1(m+1)! + \ldots + a_n(n+m)! = 0.\] This is a homogenous linear recurrence relation with constant coefficients for the factorial sequence, which is impossible unless $a_0 = a_1 = \ldots = a_n = 0$ since the factorial function grows superexponentially.  Since $\{l_k(t)\}_k$ forms a basis for $\R[t]$, if $\Phi(p(t)l_k(t)) = 0$ for all $k$ then we can conclude $p(t) = 0$.  So it is enough to show that for any nonnegative integers $i,j,m$,

\[\Phi\left( l_i(t)l_j(t) l_m(t) \right) = \Phi\left(\sum_k n_{i,j,k} l_k(t) l_m(t) \right).\]

We know that the left hand side counts the number of Carlitz arrangements of $i$ $a$'s, $j$ $b$'s, and $m$ $c$'s, while the right hand side gives the total number of pairs $(\phi, W)$ where $\phi$ is a factorization in $k$ parts with $i$ $a$'s and $j$ $b$'s with each part Carlitz, and $W$ is a Carlitz word with $k$ $x$'s and $m$ $c$'s. There is a simple bijection between these sets.  Given such a pair $(\phi, W)$, we can get a Carlitz arrangement of $i$ $a$'s, $j$ $b$'s and $m$ $c$'s by replacing the $i$th $x$ of $W$ with the $i$th part of $\phi$.  For example, if $\phi = (ab)(bab)$ and $W = cxcx$, we get the Carlitz word $cabcbab$.  This process is reversible:  given a Carlitz word on $a,b,c$ we replace the $c$'s by parentheses to make a factorization $\phi$ with only the letters $a$ and $b$, and to get $W$ we replace each maximal subword that does not contain $c$ by a single $x$, getting a word with only $c$'s and $x$'s.  For example, given the word $abcbcab$, we get the pair $W = xcxcx$, and $\phi = (ab)(b)(ab)$.  The maximality condition guarantees that $W$ will be Carlitz.\end{proof}

\begin{definition} Given a set of factorizations $A$ on an alphabet $S$, a {\it weight} is a function $w$ from $A$ and all of the restrictions of factorizations in $A$ into a polynomial ring $\R[x_1, x_2, \ldots]$ that commutes with restriction in the sense that if $\phi \in A$ and $T \subseteq S$, then $w(\phi) = w(\phi|_T) \, w(\phi|_{S\backslash T})$.
\end{definition}

Note that in particular, if $A = A_1 *A_2$ for some sets of factorizations $A_1,A_2$ then $w$ is also a weight on $A_1$ and $A_2$.  Also note that taking $T$ to be empty forces $w(\emptyset) = 1$.  Typically we will take the weight $w(\phi)$ to be a monomial $x_1^{n_1(\phi)} x_2^{n_2(\phi)} \cdots x_m^{n_m(\phi)}$ where each $n_i(\phi)$ is a statistic of some kind.  Examples include 

\begin{itemize}
\item $\len(\phi)$, the length of $\phi$
\item $\operatorname{sum}(\phi)$, the sum of the letters in $\phi$ if the symbols in $\phi$ are nonnegative integers
\item the number of distinct symbols in $\phi$
\item the number of appearances of a particular symbol
\end{itemize}

\noindent or simply $w=1$ if we wish to enumerate a finite set.  We write $\parts(\phi)$ for the number of parts of $\phi$; but $x^{\parts(\phi)}$ is in general {\it not} a weight.

\begin{definition} Let $A$ be a set of factorizations on an alphabet $S$ and $w$ be a weight on $A$.  Define the {\it Laguerre series} of $A$ with respect to $w$ to be the formal power series \[f_{A,w}(t) = \sum_{\phi \in A} w(\phi) l_{\parts(\phi)}(t)\] when this sum is well-defined as a formal power series.  For convenience we will omit the $w$ in the subscript when $w=1$, writing $f_{A,1}(t)$ as $f_A(t)$.
\end{definition}
Convergence of Laguerre series is studied in, e.g., \cite{Pollard, szasz, weniger}, where they are defined as series $$\sum_n \lambda_n^{(\alpha)}L_n^{(\alpha)}(t).$$  Here we only consider them as formal power series, we always take $\alpha = -1$, and our coefficients may be polynomials in indeterminate variables.

%In particular, we see that $l_k(t)$ is the Laguerre series for a one-letter alphabet $\{a\}$, with the only allowed factorization being $(a)(a) \cdots (a)$ with exactly $k$ parts, with respect to the weight $w=1$. 

\begin{proposition}\label{phifact}
Assume $A$ is a set of factorizations and $w$ is a weight on $A$.  Let $\Phi$ be the linear operator so that $\Phi(t^n) = n!$ and $\Phi$ fixes any other variables.  Then \[\Phi( f_{A,w}(t)) = \sum_W w(W)\] when both sides are defined, where the sum is over words $W \in A$ which are factorizations with one or no parts.
\end{proposition}

\begin{proof}
We have
%%%
\begin{align*}
\Phi(f_{A,w}(t)) &= \sum_{\phi \in A} w(\phi) \Phi( l_{\parts(\phi)}(t))\\
\end{align*}
%%%%
and $\Phi( l_{\parts(\phi)}(t))$ is $1$ when $\phi$ has one or no parts and is $0$ otherwise by~\eqref{orthogonality} since $l_0(t) = 1$ and $l_1(t)= t$.
\noindent 
\end{proof}

It is easy to see that if $A_1$ and $A_2$ are disjoint sets of factorizations on a common alphabet $S$ and $w$ is a weight on $A_1 \cup A_2$ then \[f_{A_1 \cup A_2,w}(t) = f_{A_1,w}(t) + f_{A_2,w}(t).\]  More interesting, perhaps, is the combinatorial interpretation of a product of Laguerre series.  This is our main theorem on the combinatorial properties of Laguerre series.   Recall that if $A_1,A_2$ are sets of factorizations on disjoint alphabets $S_1,S_2$, then $A_1* A_2$ was defined to be the set of factorizations $\phi$ of words on $S = S_1 \cup S_2$ so that $\phi|_{S_1} \in A_1$ and $\phi|_{S_2} \in A_2$. 

\begin{theorem} \label{key} 
Let $S_1$ and $S_2$ be disjoint alphabets with and $A_1$,$A_2$ be sets of factorizations on $S_1,S_2$ respectively, and let $w$ be a weight on $A_1 * A_2$ (and hence on $A_1$ and $A_2$.)  Then \[f_{A_1 * A_2,w}(t) = f_{A_1,w}(t)\cdot f_{A_2,w}(t).\]
\end{theorem}

\begin{proof}
By Lemma 2.3,
\begin{align*}
 f_{A_1,w}(t)\cdot f_{A_2,w}(t) &= \sum_{\phi_1 \in A_1} \sum_{\phi_2 \in A_2} w(\phi_1) w(\phi_2) l_{\parts(\phi_1)}(t) l_{\parts(\phi_1)}(t)\\
 &= \sum_{\phi_1 \in A_1, \phi_2 \in A_2, k\geq 0} n_{\parts(\phi_1), \parts(\phi_2),k} w(\phi_1) w(\phi_2) l_k(t).
\end{align*}

Fix $\phi_1 \in A_1, \phi_2 \in A_2$.  It is enough to show that $n_{\parts(\phi_1), \parts(\phi_2),k}$ is the number of factorizations with $k$ parts on $S_1 \cup S_2$ whose restrictions to $S_1$ and $S_2$ are $\phi_1$ and $\phi_2$, respectively.  Then each word of $A_1 * A_2$ will then be represented exactly once in the series $f_{A_1,w}(t)\cdot f_{A_2,w}(t)$, giving \[f_{A_1 * A_2,w}(t) = \sum_{\phi \in A_1 * A_2} w(\phi) l_{\parts(\phi)}(t) =  f_{A_1,w}(t)\cdot f_{A_2,w}(t).\]
 
For fixed $k$, we will construct a simple bijection from the set of triples $(\phi,\phi_1, \phi_2)$ where $\phi_1$, $\phi_2$ are factorizations in $A_1,A_2$ respectively and $\phi$ is a factorization on the alphabet $\{a,b\}$ with $\parts(\phi_1)$ $a$'s and $\parts(\phi_1)$ $b$'s so that each part is Carlitz, and the set of factorizations $\phi_3$ of $A_1 * A_2$ with $k$ parts.  Let $\phi_3$ be the factorization created by replacing the $n$th $a$ in $\phi$ with the $n$th part of $\phi_1$, and the $n$th $b$ with the $n$th part of $\phi_2$.  Then by construction $\phi_3 \in A_1 * A_2$: its restrictions are $\phi_1$ and $\phi_2$.  Furthermore, given an factorization $\phi_3 \in A_1 * A_2$ with $k$ parts so that $\phi_{S_1} = \phi_1$, $\phi_{S_2} = \phi_2$, we can reconstruct the factorization $\phi$ of a word on $\{a,b\}$ by replacing each subword of a factor of $\phi_3$ that uses only the letters of $S_1$, and is maximal with respect to this condition, by an $a$ and each maximal subword using only letters of $S_2$ by a $b$.  For example, if $S_1 = \{1,2\}$ and $S_2 = \{3,4\}$, with $\phi_3 = (123)(2213)(34413)$, we get the word $\phi =(ab)(ab)(bab)$.  No part of $\phi$ can have $aa$ or $bb$ by the maximality condition.  These two algorithms are inverse to each other, establishing the theorem.\end{proof}

Inductively, we see that if $A_1, \ldots, A_n$ are sets of factorizations on disjoint alphabets and $w$ is a weight on $A_1 * \cdots * A_n$, then $f_{A_1 *\cdots* A_n,w}(t) = f_{A_1,w}(t)\cdots f_{A_n,w}(t)$.  Thus we have the following extension of Lemma~\ref{linearization}:

\begin{corollary}
Let $S = \{a_1, \ldots, a_m\}$ be an alphabet of $m$ distinct letters, and $r_1, \ldots, r_m$ be fixed nonnegative integers.  Let $n_k$ denote the number of factorizations on $r_1 + \cdots + r_m$ letters in $k$ parts so that each part is Carlitz, and the $i$th letter $a_i$ of $S$ is used exactly $r_i$ times for each $i$.  Then \[\prod_{i=1}^m l_{r_i}(x) = \sum_{k\geq 0} n_k l_k(x).\]
\end{corollary}
\begin{proof}
Let $A_i$ consist only of the factorization $(a_i)(a_i) \cdots (a_i)$, with exactly $r_i$ factors.  Then $A_1 * \cdots * A_m$ consists of those factorizations for which each part is Carlitz and the letter $a_i$ is repeated $n_i$ times so we may apply Theorem~\ref{key}.  Alternately, we can adapt the proof of Lemma~\ref{linearization}.
\end{proof}

\section{Computing Laguerre series}

The Laguerre series for a set of factorizations would not be especially useful if it were difficult to compute.  Fortunately, there is an efficient method to calculate them in some situations.  It may be difficult to find a convenient formula for the coefficients of $l_k(t)$ in a given Laguerre series $f_{A,w}(t)$, but this is not needed to find an expression for $f_{A,w}(t)$.  It is enough to find the ordinary generating function.  Specifically, we define $$g_{A,w}(u) = \sum_{ \phi \in A} w(\phi) u^{\parts(k)}.$$  If a nice form of $g_{A,w}(u)$ is known, we may obtain the Laguerre series $f_{A,w}(t)$ by applying the linear transformation $T$ that sends $u^k$ to $l_k(t)$.  As it happens, $T$ can be computed in many situations using the inverse Laplace transform.  We have \[ \mathcal{L}\{l_k(t)\} = \frac{1}{s(1-s)} \left( \frac{1-s}{s}\right)^k\] for $k\geq 1$, where $\mathcal{L}$ is the Laplace transform; this is easily proved from the formula for $l_k(t)$ in terms of the generalized Laguerre polynomials, the fact that $\mathcal{L}\{ t^i\} = \frac{i!}{s^{i+1}}$, and the binomial theorem.
 
Therefore, if $g_{A,w}(0) = 0$, we have by linearity \[ T\{g_{A,w}(u)\} = f_{A,w}(t) = \mathcal{L}^{-1}\left\{ \frac{g_{A,w}(\frac{1-s}{s})}{s(1-s)}\right\}\] when the right-hand side is well-defined.  If $g_{A,w}(0) \neq 0$, we can calculate  \[ f_{A,w}(t) = T\{g_{A,w}(u) - g_{A,w}(0)\} + g_{A,w}(0)  =  \mathcal{L}^{-1}\left\{ \frac{g_{A,w}(\frac{1-s}{s}) - g_{A,w}(0)}{s(1-s)}\right\} + g_{A,w}(0)\] since $l_0(t) = 1$.  The use of the inverse Laplace transform here is not central to the theory, but it is convenient since many software packages provide symbolic calculation of the inverse Laplace transform, making it easy to implement the transformation $T$.

In many cases, we can use standard generating function techniques to calculate $g_{A,w}(u)$.  For example, let $A$ be the set of all factorizations on a single-letter alphabet, and let $w$ be the weight $w(\phi)= x^{\len(\phi)}$.  Then we see that \[g_{A,w}(u) = \sum_{k=0}^\infty u^k(x+x^2 + \ldots)^k\] since we choose a factorization $\phi$ by choosing the number of parts and then choosing the size of each part.  We may simplify this expression to get $g_{A,w}(u) = \frac{1-x}{1 - x - ux},$ and calculate  $f_{A,w}(t) = T\{g_{A,w}(u)\} = e^{tx}$.  Taking the coefficient of $x^k$ shows that the Laguerre series for all factorizations on this single-letter alphabet with length $k$ is $\frac{t^k}{k!}$ with respect to the weight $w=1$.  Thus Proposition~\ref{phifact} and Theorem~\ref{key} tell us that \[\Phi\left(\prod_{k=1}^n \sum_{i=0}^{r_k} \frac{t^i}{i!}\right)\] is the total number of words on an alphabet $S = \{c_1, \ldots, c_n\}$ so that $c_k$ is used at most $r_k$ times, where $r_1, \ldots, r_n$ is a set of nonnegative integers.  Of course, this formula can also be seen directly.

Laguerre series are sometimes useful for counting words with certain subwords prohibited.  Consider the following problem: given a word $W$ and a subword $W'$ of $W$, how many words are there avoiding $W'$ as a subword that can be made from $W$ using each symbol at most as many times as it appears in $W$?   To answer this, we form the set $A_1$ of factorizations that can be made from $W$ that use only letters from $W'$ and do not contain $W'$ as a subword of any factor.  Letting $A_2$ be the set of all factorizations of words that can be made from $W$ in the above sense that do not use any letters from $W'$, we see that $A_1 * A_2$ is the set of all factorizations of words that can be made from $W$ so that no part contains $W'$ as a subword.  Setting $w=1$, we may perform a brute force calculation to compute the Laguerre series $f_{A_1}(t)$.  Since there are no subword restrictions on the remaining letters, we have \[f_{A_2}(t) = \prod_c \sum_{i=0}^{n_c} \frac{t^i}{i!}\] where the product is over the letters $c$ used in $W$ that are not used in $W'$, and $n_c$ is the number of times $c$ is used in $W$.  So the solution is given by $\Phi(f_{A_1}(t)f_{A_2}(t))$.

This method is practical if there are not too many letters used in the $W'$, and the number of times these letters are repeated in $W$ is small.  For example, how many words are there that can be made from the letters in ``CONSTANTINOPLE'' that avoids the subword ``TNT''?  We focus on the letters $T$ and $N$, forming the set $A_1$ consisting of those factorization of words using only these letters, with three or fewer $N$'s and two or fewer $T$'s, so that no factor contains ``TNT''.  We then perform a brute force count using a programming language such as Sage to explicitly calculate the Laguerre series for $A_1$ with respect to $w=1$, getting \[f_{A_1}(t) = \frac{1}{12}t^5 + \frac{5}{12}t^4 + \frac{2}{3} t^3 + t^2 + t + 1.\]  Then forming $A_2$ as above, the set of factorizations of any words made from the remaining letters ``COSAIOPLE'', we calculate $f_{A_2}(t) = (t + 1)^7(\frac{1}{2}t^2 + t +1)$ since all of the letters appear only once, except O which appears twice.  Therefore, the answer is \[\int_0^\infty e^{-t} \left(\frac{1}{12}t^5 + \frac{5}{12}t^4 + \frac{2}{3} t^3 + t^2 + t + 1\right)\left(t +1\right)^7\left(\frac{1}{2}t^2 +t +1\right)\, dt = 9{,}854{,}474{,}467.\]

A more general method for the subword avoidance problem is the cluster method developed by Goulden and Jackson \cite{Goulden-Jackson1,Goulden-Jackson2}.  This a powerful tool using linear algebra and inclusion-exclusion to find rational generating functions for words avoiding any given set of subwords. Noonan and Zeilberger \cite{Doron1} give generalizations and a good introduction to the idea; Edlin and Zeilberger also give an extension to cyclic words \cite{Doron2}.

Now consider the problem of counting words that have no subword consisting of $m$ identical letters.  These are words that avoid the subword pattern $1^m$, and are sometimes called $m$-Carlitz words; when $m=2$ we have the ordinary Carlitz words.  To find the generating function, let $A$ be the set of factorizations on a one-letter alphabet with each part having length smaller than $m$, and again let $w(\phi)= x^{\len(\phi)}$.   We see that \[g_{A,w}(u) = \sum_{n = 0}^\infty u^n(x + \ldots + x^{m-1})^n =  \frac{1 - x}{1 -x - u(x - x^m)}\] and so we compute 
\begin{align}\label{lacefunforsmallruns}
f_{A,w}(t) = T\{g_{A,w}(u)\} = \exp\left(t \cdot\frac{x - x^m}{1 - x^m}\right).
\end{align} 

Taking the coefficient of $x^n$ in~\eqref{lacefunforsmallruns} gives the Laguerre series for the set of factorizations with length $n$ so that each part is smaller than $m$.  This gives a generalization of Theorem~\ref{fundamental}.  If $m_1, \ldots, m_k, n_1, \ldots, n_k$ are nonnegative integers, and $p_{m,n}(t)$ are polynomials defined by $\sum_{n=0}^\infty p_{m,n}(t) \,x^n = \exp{\left( \frac{t(x - x^m)}{1- x^m}\right)}$, we see that \[\Phi\left( \prod_{i =1}^k p_{m_i,n_i}(t)\right)\] is the total number of $k$-ary words that use the letter $i$ exactly $n_i$ times and do not contain the subwords $i^{m_i}$. Thus the number of arrangements of the word ``WALLAWALLA'' with no LLL, AAA or WW as consecutive subwords is \begin{align*}
\int_0^\infty e^{-t} p_{3,4}(t)\cdot p_{3,4}(t) \cdot p_{2,2}(t)\, dt &= \int_0^\infty e^{-t}\left(\frac{1}{24}t^4 - t^2 + t\right)^2 \cdot \left(\frac{1}{2}t^2 - t\right) \, dt\\
&= 1584.
\end{align*}

Recalling again the formula 
%%%
\begin{align}\label{phirule}
\Phi(e^{tf}) = \frac{1}{1-f},
\end{align}
%%%
we see that the generating function for the number of $k$-ary $m$-Carlitz words of length $n$ is given by \[\Phi\left(\exp\left( kt \cdot\frac{ x - x^m}{1-x^m} \right)\right) = \frac{1 - x^m}{1 - kx - (k-1)x^m}.\]  Another derivation of this formula is given by Burstein and Mansour \cite[Example~2.2]{Mansour2}.

We may set indeterminates to count the number of occurrences of each symbol. Let $w(\phi) = x_1^{i_1} x_2^{i_2} \cdots x_k^{i_k}$, where $i_j$ is the number of appearances of $j$ in a factorization $\phi$ on the alphabet $[k]$. By~\eqref{lacefunforsmallruns} and~\eqref{phirule}, we see that \begin{align*}\displaystyle\Phi\left( \prod_{i=1}^k  \exp\left(t \cdot\frac{x_i - x_i^m}{1-x_i^m}\right) \right) = \frac{1}{1 - \sum_{i=1}^k \frac{x_i - x_i^m}{1 - x_i^m}}\end{align*} gives the sum of all weights in $m$-Carlitz $k$-ary words.  This can also be found by substituting $x_i + x_i^2 + \ldots +x_i^{m-1}$ for $x_i$ in Lemma~\ref{carlitzgf}.

\section{Vincular patterns}

We will show how to use the techniques of Sections 2 and 3 to find generating functions for the number of words avoiding vincular patterns with only ones.  In the introduction we defined what it means for a word $W$ to contain or avoid a vincular pattern $\tau = 1^{m_1}\d 1^{m_2}\d\ldots\d 1^{m_n}$, but in order to make use of Theorem~\ref{key} we will extend this definition to factorizations on a single-letter alphabet.  We say that a factorization $\phi$ on some singleton alphabet $\{c\}$ contains $\tau$ if every word $W$ with $W|_{\{c\}} = \phi$ contains $\tau$.  Otherwise we say that $\phi$ avoids $\tau$.  For example, if $\tau = 111\d 11 \d 11$ then $\phi = (aaaaa)(a)(aaaa)$ contains $\tau$ since every word $W$ with $W|_{\{a\}} = \phi$ contains disjoints runs of letters $aaa$, $aa$ and $aa$, in that order.  Note that the first two runs $aaa$, $aa$ are contained in the larger run $aaaaa$ - words separated by dashes in a vincular pattern are not required to be separated in the word $W$.  However, the factorization $\phi = (aaaa)(aaa)$ avoids $\tau$ since the word $W = aaaabaaa$ has $W|_{\{a\}} = \phi$ and $W$ avoids $\tau$.

An equivalent formulation may be useful. Suppose $\tau = 1^{m_1}\d 1^{m_2}\d\ldots\d 1^{m_n}$ and $\phi$ is a factorization $\phi = (c^{l_1})(c^{l_2})\cdots (c^{l_p})$ on a singleton alphabet $\{c\}$.  Then $\phi$ contains $\tau$ if and only if there are integers $1 \leq i_1 < i_2 < \cdots < i_r < n$ and $1 \leq j_1 < \cdots < j_r \leq p$  so that
%%%
\begin{alignat*}{7}
m_1    &&\; + \cdots + \;&& m_{i_1} &&\; \leq \;&&&  l_{j_1} \\
m_{i_1 +1}  &&\; + \cdots + \;&&  m_{i_2} &&\; \leq \;&&& l_{j_2} \\
\vdots\;\;\; &&                 && \vdots\;\;\; &&     &&& \;\vdots \\
m_{i_r +1}  &&\; + \cdots + \;&& m_n &&\; \leq \;&&& l_{j_r}.
\end{alignat*}
%%%
%\begin{align*}
%m_1 + \cdots + m_{i_1} &\leq l_{j_1}\\
%m_{i_1 +1} + \cdots + m_{i_2} &\leq l_{j_2}\\
%\vdots
%m_{i_r +1} + \cdots + m_n &\leq l_{j_r}.\\
%\end{align*}
%%%
For example, if $\tau = 111\d 11 \d 11$ as above then the factorization $\phi = (aaaaa)(a)(aaaa)$ contains $\tau$ since $3+2 \leq 5$ and $2 \leq 4$.

\begin{theorem}{\label{vincgf}}  Let $m_1, \ldots, m_n$ be positive integers, and let $A$ be the set of $k$-ary words avoiding the pattern $\tau = 1^{m_1} \d 1^{m_2} \d \cdots \d 1^{m_n}$.  Then \[ \sum_{a_1 \cdots a_l \in A} x_{a_1} x_{a_2} \cdots x_{a_l} =  \Phi\left(   \prod_{i=1}^k\left[ e^{tx_i} - T\left\{G_\tau(x_i, u)\right\}\right]\right)\] where $T$ is the operator defined in Section 3, and 
\begin{align}\label{G}
G_\tau(x,u) =\frac{ ux^{m_1}(1-x) }{(1-x - u(x-x^{m_1}))(1-x- ux)} \prod_{i=2}^n \left[ x^{m_i} + \frac{ux^{m_i}(1 - x^{m_i})}{1-x - u(x-x^{m_i})}  \right].
\end{align}
\end{theorem}
\begin{proof}
Earlier we observed that $e^{tx}$ is the Laguerre series for all factorizations using only a single letter with respect to the weight $w(\phi) = x^{\len(\phi)}$.  So it is enough to show that $G_\tau(x,u) = g_{A,w}(u)$ where $A$ is the set of factorizations on some single letter alphabet containing the given vincular pattern; the difference will then count the factorizations on a single letter which avoid the pattern.  For the remainder of the proof, we will use only the alphabet $\{1\}$.
We will use repeatedly the following basic facts:

\begin{fact}  Given two sets of factorizations $A_1$ and $A_2$, let $$A_3 = \{ (\phi_1)\ldots(\phi_l)(\psi_1)\ldots(\psi_m): \phi = (\phi_1)\ldots(\phi_l) \in A_1, \psi = (\psi_1)\ldots(\psi_m) \in A_2\}.$$  Then $g_{A_3,w}(u) = g_{A_1,w}(u) \cdot g_{A_2,w}(u)$ when both sides are defined.
\end{fact}

\begin{fact}Let $A$ be a set of factorizations on the alphabet $\{1\}$.  Then for a fixed integer $k\geq 0$, let $A_k$ be the set consisting of each factorization of $\phi$ with $k$ ones appended to the last factor: $A_k = \{(\phi_1)(\phi_2) \cdots (\phi_n 1^k): (\phi_1) \cdots (\phi_n) \in A\}$.  Then $g_{A_k,w}(u) = x^kg_{A,w}(u)$.\end{fact}

We say that a factorization {\it minimally} contains the pattern $1^{m_1} \d 1^{m_2} \d \cdots \d 1^{m_n}$ if it contains the pattern, but no longer contains the pattern if the last letter of the last factor is removed.  For example, $(11)(1)(1)(111)$ contains $111$ minimally, contains $11\d11$ but not minimally, and does not contain $1111$ at all.  Define a {\it truncation} of a factorization $\phi = (\phi_1) \cdots (\phi_n)$ to be a factorization $\hat{\phi} = (\phi_1) \cdots (\phi_{i-1}) (\hat{\phi_i})$, where $1 \leq i \leq n$ and $\hat{\phi_i}$ is a nonempty initial substring of $\phi_i$.  Any factorization containing a pattern has a unique truncation that contains that pattern minimally, so we will count the factorizations containing it minimally and then multiply by the appropriate expression to count all factorizations that contain the pattern.

First, suppose the pattern is just $1^{m_1}$.  A factorization minimally containing this pattern is any factorization whose last part has size $m_1$, and all other parts have size strictly smaller.  So the generating function for these factorizations is \[ \sum_{n=0}^\infty (u(x + \ldots + x^{m_1-1}))^nux^{m_1} = \left(\frac{ 1 - x}{1 - x - u(x - x^{m_1})}\right)ux^{m_1}.\]

Now let the generating function for the number of words minimally containing the pattern $1^{m_1} \d \cdots \d 1^{m_{n}}$ be $g_n(x,u)$; we will find $g_n(x,u)$ in terms of $g_{n-1}(x,u)$.  Let $\phi$ be a factorization minimally containing $1^{m_1} \d\cdots\d 1^{m_{n-1}} \d 1^{m_n}$; then $\phi$ has a unique truncation $\hat{\phi}$ that minimally contains  $1^{m_1} \d\cdots \d 1^{m_{n-1}}$. Thus to find $g_n(x,u)$ we multiply $g_{n-1}(x,u)$ by the appropriate factor, mimicking the following process by which any such $\phi$ is attained uniquely from some $\hat{\phi}$. If $\phi$ and $\hat{\phi}$ have the same number of factors, $\phi$ must be $\hat{\phi}$ with exactly $m_n$ $1$'s appended to the last factor; so we multiply by $x^{m_n}$.  Otherwise, we can assume a new factor is necessary.  In this case, we may lengthen the final part of $\phi$ by adding $1$'s, but to avoid having the pattern $1^{m_1} \d \cdots \d 1^{m_{n}}$ without adding any factors at most $m_n-1$ should be added.  This means we should first multiply by $$1 + x + \ldots + x^{m_n-1} = \frac{1-x^{m_n}}{1-x}.$$  Then, we may append some (possibly empty) sequence of factors to $\hat{\phi}$, each having length less than $m_n$.  This amounts to multiplying by $$\sum_{i=0}^\infty (u(x + \ldots + x^{m_n - 1}))^i = \frac{ 1 - x}{1 - x - u(x - x^{m_{n}})}.$$ Finally, we append a new factor which is just $1^{m_n}$, multiplying by $ux^{m_n}$.   Thus we multiply by \[\frac{1-x^{m_n}}{1-x} \cdot \frac{ 1 - x}{1 - x - u(x - x^{m_{n}})}\cdot ux^{m_n}.  \]  So the ordinary generating function minimally avoiding $1^{m_1} \d\ldots \d 1^{m_{n}}$ is exactly \[g_n(x,u) = g_{n-1}(x,u) \cdot \left(x^{m_n} + \frac{ ux^{m_n}(1 - x^{m_n})}{1 - x - u(x - x^{m_n})}\right).\]  Finally, once we have a factorization minimally containing the pattern, we may add any number of ones to the last factor and append any list of factors and the resulting factorization will still contain the pattern.  This amounts to multiplying by \[   \sum_{j=0}^\infty\sum_{i=0}^\infty x^j \left(u( x + x^2 + \ldots)\right)^i = \frac{1}{1-x} \frac{1 -x}{1 - x - ux}.\] Putting this all together, we see inductively that the generating function for factorizations containing the pattern is
\begin{align*}
\frac{ ux^{m_1}(1-x) }{(1-x - u(x-x^{m_i}))} \prod_{i=2}^n \left[ x^{m_i} + \frac{ux^{m_n}(1 - x^{m_i})}{1-x - u(x-x^{m_i})}\right] \frac{1}{1 - x - ux}.
\end{align*}
\end{proof}

\section{Cyclically avoiding patterns}

We say that a word $W$ cyclically avoids a vincular pattern $\tau$ if $W$ avoids $\tau$ no matter how its letters are cycled.  More formally, let $r$ be the function that cycles $W$, moving the last letter into the first position: $r(a_1 \cdots a_n) = a_n a_1 \cdots a_{n-1} $.  Then $W$ cyclically avoids $\tau$ if $r^k(W)$ avoids $\tau$ for each $k$.  Note that we are not considering words to be the same up to rotation here; $W$ and $r(W)$ are considered to be different words.

Let $\tau = 1^m \d 1^m \d \cdots \d 1^m$ be the pattern consisting of $n$ copies of $1^m$ separated by dashes.  In order to find the generating function for the number of words cyclically avoiding the pattern $\tau$, we will need a little more information than provided by the generating function $G_\tau(x,u)$ defined by ~\eqref{G}.  Let $H(x,u,v) = g_{A,w}(u)$ where $A$ is the set of factorizations on the alphabet $\{1\}$ {\it avoiding} the pattern $\tau$ and $\hat{w}$ is the weight $\hat{w}(\phi) = x^{\len(\phi)} u^{\parts(\phi)}  v^{ \fst(\phi)}$ where $\fst(\phi)$ is the size of the first factor of $\phi$.  Note that $\hat{w}$ is trivially a weight by our definition since we are using a singleton alphabet, but generally is not. We will find a  closed-form expression for $H(x,u,v)$, although it is rather unwieldy.

\begin{lemma}\label{hlemma} The generating function $H(x,u,v)$ is given by
\begin{align}
H(x,u,v) = 1 &+ \left[ \frac{1-x}{(1-vx)(1-x-ux)}\right] \biggl[{u(vx - (vx)^{mn})} +\biggl.\notag\\
& \left.\frac{{u^{2}x^{m} \left({\left(1-vx\right)} {\left(z - (vx)^{m}\right)} z^{n - 1} - {\left(1-(vx)^{m}\right)} {\left(z^n -(vx)^{m n}\right)}\right)} }{{\left(z-(vx)^{m}\right)} {\left({1-x - u\left(x - x^{m}\right)}\right)}}\right]\label{H}
\end{align}
where $$z = x^m + \frac{ux^m(1 - x^m)}{1 - x - u(x - x^m)}.$$   
\end{lemma}
\begin{proof}

Fix $n>0$, and let $\tau_n$ be the pattern $1^m \d 1^m \d \cdots \d 1^m$ with $n-1$ dashes.  Then by the proof of Theorem~\ref{vincgf}, we know that if $G_n(x,u)$ is the generating function~\eqref{G} for the number of factorizations on the alphabet $\{1\}$ containing $\tau_n$, then $$G_n(x,u) = \frac{ ux^{m}(1-x) }{(1-x - u(x-x^{m}))(1-x- ux)} \left(  x^{m} + \frac{ux^{m}(1 - x^{m})}{1-x - u(x-x^{m})}\right)^{n-1}$$  since in this case $m_1 = m_2 = \ldots = m.$  We have different cases depending on how much of the pattern $\tau_n$ is contained in the first part of a factorization.  Let $A$ be the set of all factorizations on the alphabet $\{1\}$ that contain $\tau_n$, and define $A_0, A_1, \ldots, A_n$ by 
\begin{align*}
A_0 &= \{\phi =  (\phi_1)\cdots(\phi_l) \in A : \phi_1\text{ avoids }\tau_1 = 1^m\}\\
A_i &= \{\phi = (\phi_1)\cdots(\phi_l)\in A: \phi_1 \text{ contains }\tau_i \text{ but }\phi_1 \text{ avoids } \tau_{i+1}\} \text{  for }1 \leq i \leq n-1\\
A_n &= \{\phi = (\phi_1)\cdots(\phi_l)\in A : \phi_1 \text{ contains }\tau_n \}.
\end{align*}

First, suppose that $\phi \in A_0$.  Then the first part of $\phi$ is one of $1, 11, \ldots, 1^{m-1}$, and the factorization $(\phi_2)\cdots(\phi_k)$ of the remaining parts can be any factorization containing the pattern.  With the exponent of $v$ recording the size of the first factor, we see that the generating function corresponding to this case is  
\begin{align*}
g_{A_0,w}(u) &= u(vx +(vx)^2 + \ldots + (vx)^{m-1}) G_n(x,u) \\
&= u\cdot \frac{vx - (vx)^m}{1 - vx}  G_n(x,u).
\end{align*}

Now suppose that $\phi \in A_i$, $1 \leq i \leq n-1$. Then the first part of $\phi$ is one of $1^{im}, 1^{im+1}, \ldots, 1^{im + m-1}$ and $(\phi_2)\cdots(\phi_l)$ can be any factorization containing $\tau_{n-i}$.  So the generating function corresponding to $A_i$ is  
\begin{align*}
g_{A_i,w}(u) &= u((vx)^{im} + (vx)^{im+1} + \ldots + (vx)^{im + m -1}) G_{\tau_{n-i}}(x,u)\\
&=  u(vx)^{im}\frac{1 - (vx)^{m}  }{1 - vx}G_{n-i}(x,u).
\end{align*} 
If $\phi \in A_n$, then the first factor of $\phi$ can have any length that is $nm$ or more, and the remaining parts of $\phi$ can be any factorization.  Therefore 
\begin{align*}
g_{A_n,w}(u) &=  u((vx)^{nm} + (vx)^{nm+1} + \ldots)\sum_j(u(x +x^2 + \ldots))^j\\
&= \frac{  u(vx)^{nm}}{(1 - vx)(1 - \frac{ux}{1-x})}.
\end{align*}
To find the generating function for factorizations avoiding the pattern, we must subtract $g_{A_i,w}(u)$, $0 \leq i \leq n$, from the generating function for all one-letter factorizations, which is 
\begin{align*}
1 + u(vx + (vx)^2 + \ldots)\sum_{k=0}^\infty(u(x + x^2 + \ldots))^k = 1 + \frac{uvx}{(1-vx)(1 - \frac{ux}{1-x})}.\\
\end{align*}
We have
\begin{align*}
H(x,u,v) =&\, 1 + \frac{uvx}{(1-vx)(1 - \frac{ux}{1-x})} - \sum_{i=0}^n g_{A_i,w}(u)\\
=& \,1 + \frac{uvx}{(1-vx)(1 - \frac{ux}{1-x})} -  u\cdot \frac{vx - (vx)^m}{1 - vx}  G_{n}(x,u)- \frac{  u(vx)^{nm}}{(1 - vx)(1 - \frac{ux}{1-x})} \\
& -  \sum_{i=1}^{n-1} u(vx)^{im}\frac{1 - (vx)^{i}  }{1 - vx}G_{n-i}(x,u).
\end{align*}
After simplifying, we have the desired formula.
\end{proof}

\begin{theorem}\label{cyclicvinc}
Let $H(x,u,v)$ be the generating function~\eqref{H}.  Let $A$ be the set of words cyclically avoiding the pattern $\tau = 1^m \d 1^m \d\cdots \d 1^m$, with $n-1$ dashes, and let $w$ be the weight on $k$-ary words with $w(s_1 \ldots s_l) = x_{s_1} x_{s_2} \cdots x_{s_l}$.  Then the generating function $\sum_{W\in A} w(W)$ is given by

\[1 + \sum_{i=1}^k \Phi\biggl( t^{-1}\cdot T\left\{ u \frac{d^2}{dv\, du}\bigg|_{v=1}   H(x_i,u,v)\right\} \biggl(- 1 +\prod^k_{\substack{j=1\\
 j\neq i}} T\{H(x_j,u,1)\}\biggr)\biggr) + \sum_{i=1}^k \frac{x_i - x_i^{mn}}{1-x}.\]
In particular, letting $x_i = x$ for each $i$ gives:
\[\sum_{W \in A} x^{\len(W)} = 1 + k\cdot \Phi\biggl( t^{-1}\cdot T\left\{ u \frac{d^2}{dv\, du}\bigg|_{v=1}   H(x,u,v)\} \big(T\{H(x,u,1)\}\right)^{k-1}-1\big)\biggr) + \frac{k(x - x^{mn})}{1-x}.\]
\end{theorem}
\begin{proof}

We will count all words which avoid $\tau$ and begin with $i$, where $1 \leq i \leq k$, and then sum over $i$.  Let $\mathcal{W}$ be the set of all $k$-ary words $W$ that begin with $i$, do not end in $i$, and avoid the pattern $\tau$.  Any $W \in \mathcal{W}$ will in fact {\it cyclically} avoid $\tau$.  To see this, it is enough to note that the restriction $r^p(W)|_{\{j\}}$, for any $j,p$, will avoid the pattern.  Such a factorization will either be a permutation of the parts of $(W)|_{\{j\}}$, or will have some part of $(W)|_{\{j\}}$ divided into the first and last part of $r^p(W)|_{\{j\}}$. Permuting the factors will not cause $(W)|_{\{j\}}$ to contain the pattern by the symmetry of $\tau$, and splitting a part into two cannot create an instance of the pattern either.

We can characterize $\mathcal{W}$ as follows.  Suppose $\phi = (\phi_1)\ldots(\phi_p)$ is a factorization over the alphabet $[k]\backslash\{i\}$.  Then we can insert $i$'s appropriately in between factors so that the result starts with $i$ and avoids the pattern.  Given a factorization $\psi = (\psi_1)\ldots(\psi_p)$ which avoids the pattern and has the same number of parts as $\phi$ but only uses the symbol $i$, let $W_{\phi,\psi} = \psi_1 \phi_1 \psi_2 \phi_2\ldots \psi_p \phi_p$ be the concatenation of the alternating factors of $\phi$ and $\psi$.  Then $W_{\phi,\psi}$ starts with $i$, will not end in $i$, and avoids $\tau$, so $W_{\phi,\psi} \in \mathcal{W}$.  Furthermore, if $W \in \mathcal{W}$, let $\phi$ be the restriction $W|_{[k]\backslash \{i\}}$, and $\psi = W|_{\{i\}}$.  Then $\phi$ and $\psi$ will have the same number of factors, and $W = W_{\phi,\psi}$.  So we see that $\mathcal{W}$ coincides with the set of $W_{\phi,\psi}$ where $\phi,\psi$ are nonempty factorizations over the alphabets $[k]\backslash \{i\}$ and $\{i\}$, respectively, which avoid $\tau$ and have the same number of parts.
 
The words $W \in \mathcal{W}$ we have described always end in a letter besides $i$; to get all words avoiding the pattern starting with $i$, we associate $W\in \mathcal{W}$ with a class $$\{W, r^{-1}W, r^{-2}W, \ldots,r^{-B(W)+1}W\},$$ where $B(W)$ is the number of adjacent $i$'s at the beginning of $W$. These rotated words will also cyclically avoid the pattern, as we have explained, and begin with at least one $i$.  Furthermore, any word $V$ beginning with $i$ and avoiding the pattern will be in the class of a unique word $W$: let $W = r^CV$ where $C$ is the number of $i$'s on the end of $V$, if any.  Therefore we will count all words $W$ that avoid the pattern and begin with $i$, but do not end with $i$, with the multiplicity $B(W)$, the number of $i$'s at the beginning of the word $W$.  This will account for all words beginning with $i$ that use at least one letter $j \neq i$.  So the weight of all $k$-ary words that begin with $i$, cyclically avoid the pattern $\tau$, and contain at least one letter $j \neq i$ is $$\sum_{W \in \mathcal{W}} B(W)w(W) = \sum_{\psi,\phi} \fst(\phi) w(\phi)w(\psi)$$ where the sum on the right-hand side is taken over all factorizations $\phi,\psi$ over the alphabets $[k]\backslash \{i\}$ and $\{i\}$, respectively, which avoid $\tau$ and have the same number of parts.

For any $1 \leq j \leq k$ let $A_j$ be the set of nonempty factorizations on the alphabet $\{j\}$ that avoid the pattern. Define
\[b_p = \sum_{{\substack{\phi \in A_1 * \cdots *\hat{A_i} * \cdots * A_k\\\parts(\phi) = p}}} w(\phi),\] where the $\hat{A_i}$ indicates that $A_i$ is removed, and let \[a_p = \sum_{\phi \in A_i, \parts(\phi) = p} \fst(\phi)x_i^{\len(\phi)}.\]  Then by the previous discussion, we see that $$\sum_{W \in \mathcal{W}} B(W) w(W) = \sum_{p= 1}^\infty a_p b_p$$ is the weight of all $k$-ary words avoiding the pattern $\tau$ which begin with $i$ and use at least one letter $j \neq i$.  It remains to evaluate the right-hand side.

We make use of the well-known fact that the generalized Laguerre polynomials obey the orthogonality relation
\begin{align}\label{lagortho}
\int_0^\infty x^\alpha e^{-x} L_i^{(\alpha)}(x) L_j^{(\alpha)}(x) \, dx = \frac{\Gamma(i + \alpha + 1)}{i!} \delta_{i,j}.  
\end{align}
which can be found in, e.g., \cite[p. 241]{ortho}.  Although this only holds in general when $\alpha > - 1$, it is true for $\alpha = -1$ if $i \geq 1$.   We have
\begin{align}\label{mylagortho}
\Phi(it^{-1} l_i(t) l_j(t)) = \delta_{i,j}
\end{align}
for $i\geq 1$. Thus for a function $f(t)$ that is a formal sum of the Laguerre polynomials $l_p(t)$, we can extract the coefficient of $l_p(t)$ by evaluating $\Phi( pt^{-1} l_p(t) f(t))$, for $p\geq 1$.  Furthermore, we see that if $f(t) = \sum_{p\geq 1} pa_p l_p(t)$ and $g(t) = \sum_{p\geq 1} b_p l_p(t)$, we have
\begin{align}\label{ab}
\Phi(t^{-1} f(t)g(t)) = \sum_{p\geq 1} a_p b_p
\end{align}
when both sides are well-defined.  We require $p\geq 1$ since the orthogonality relation~\eqref{mylagortho} fails when $i = 0$.

In this case, $\sum_{p \geq 1} b_p l_p(t)$ is the Laguerre series for {\it nonempty} $k$-ary words avoiding the pattern that do not use $i$; it is given by $\left(\prod_{j \neq i} T\{H(x_j,u,1)\}\right) - 1$.  To find $\sum_{p \geq 1} pa_p l_p(t)$, recall from the Lemma~\ref{hlemma} that $H(x,u,v) = \sum_{\phi \in A_i} x^{\len(\phi)} u^{\parts(\phi)}  v^ {\fst(\phi)}$.  So \[u \frac{d^2}{dv\, du}\bigg|_{v=1}   g(x_i,u,v)  =   \sum_{\phi \in A_i, \phi \neq \emptyset}  \fst(\phi)\cdot \parts(\phi) \cdot x_i^{\len(\phi)} u^{\parts(\phi)}.\] Then applying $T$, we see that \[T\left\{u \frac{d^2}{dv\, du}\bigg|_{v=1}H(x_i,u,v)\right\} = \sum_{p \geq 1} p a_p l_p(t).\]  Thus by~\eqref{ab},
\begin{align}\label{beginningwithi}
\Phi\left(t^{-1}\cdot T\left\{  u\frac{d^2}{dv\, du}\bigg|_{v=1}   H(x_i,u,v)\right\}\cdot \prod_{j \neq i} \bigl(T\{G(x_j,u)\}\bigr)\right)
\end{align}
is the generating function for the number of $k$-ary words that begin with $i$, use at least one $j\neq i$, and cyclically avoid the pattern.  This does not account for the words that use only the letter $i$ and avoid the pattern, which are the words $i^k$, with $k< mn$.  So we must add $x_i + x_i^2 + \ldots + x_i^{mn-1} = \frac{x_i - x_i^{mn}}{1 - x_i}$ to ~\eqref{beginningwithi} to get the full generating function for all words that begin with $i$ and cyclically avoid $\tau$.  Summing over $i$ and adding $1$ to represent the empty factorization gives \[1 + \sum_{i=1}^k \Phi\biggl( t^{-1}\cdot T\left\{ u \frac{d^2}{dv\, du}\bigg|_{v=1}   H(x_i,u,v)\right\} \biggl(- 1 +\prod^k_{\substack{j=1\\
 j\neq i}} T\{H(x_j,u,1)\}\biggr)\biggr) + \sum_{i=1}^k \frac{x_i - x_i^{mn}}{1-x}\] as desired.
\end{proof}

If we set $n=1$, considering $k$-ary words that cyclically avoid $1^m$, the formula simplifies considerably.  After some computation, which we omit here, we arrive at the following formula.

\begin{corollary}
Let $A$ be the set of nonempty $k$-ary words avoiding $1^m$.  Let $w$ be the weight $w(a_1 \cdots a_l) = x_{a_1} x_{a_2} \cdots x_{a_l}$.  Then 
\[\sum_{W \in A} w(W) = \sum_{i = 1}^k \frac{  x_i^{2m} - m x_i^{m + 1} + (m - 1)x_i^{m}}{ (x_i^{m} - 1)(x_i -1) } + \frac{\sum_{i=1}^k \frac{ (m - 1)x_i^{m+1} - m x_i^{m} + x_i}{(x_i^{m} - 1)^2} }{1 - \sum_{i =1}^k \frac{x_i^{m} - {x_i}}{ x_i^{m} - 1}}.\]
\end{corollary}

In the book by Heubach and Mansour \cite{Mansour1}, the authors define a cyclic Carlitz composition as a Carlitz composition so that the first and last parts are not equal; they ask (Research Direction 3.3) for the generating function for the number of cyclic Carlitz compositions.  If we let $k$ approach infinity, $m=2$, and $x_i = x^i$, we get the following.

\begin{corollary}
Let $A$ be the set of cyclic Carlitz compositions.  Then
\[ \sum_{W \in A} x^{\operatorname{sum}(W)} = \frac{\sum_{i=1}^\infty \frac{x^i}{(1 + x^i)^2}}{1 - \sum_{i=1}^\infty \frac{x^i}{1 + x^i}} + \sum_{i=1}^\infty \frac{x^{2i}}{1 + x^i}.\]
\end{corollary}

Setting $x_i = x$ in Corollary 5.3 and simplifying gives the following formula.

\begin{corollary}
Let $A$ be the set of nonempty $k$-ary words that cyclically avoid $1^m$.  Then \[\sum_{W \in A} x^{\len(W)} = \frac{1 - x^{m-1}}{ 1 - x} \left(kx + (k-1)x\left(\frac{m - (m-1)kx}{1 - kx + (k-1)x^m} - \frac{m}{1 - x^{m}}\right)\right).\]
\end{corollary}

This was found by Burstein and Wilf \cite{Wilf}. They go on to show that the number of $k$-ary words of length $n$ cyclically avoiding $1^m$ is asymptotically $\beta^n$, where $\beta$ is the positive root of $x^{m+1} = (k - 1)(1 + x + x^2 + \ldots + x^m)$. In fact, they extract an explicit formula when $n$ is sufficiently large.

We can also give a cyclic version of Theorem~\ref{fundamental}, which can be derived through a similar argument to the proof of Theorem 5.2.

\begin{corollary}
Let $n_1, \ldots, n_k$ be positive integers, and let $p_{m ,n}(t)$ be defined as before by $\sum_{n=0}^\infty p_{m,n}(t) \,x^n = \exp{\left( \frac{t(x - x^m)}{1- x^m}\right)}$.  Then \[N\cdot \Phi\left(t^{-1}\cdot \prod_{i =1}^k p_{m,n_i}(t)\right)\] is the total number of $k$-ary words that use the letter $i$ exactly $n_i$ times and cyclically avoid $1^m$, where $N = \sum_{i=1}^k n_i$ is the total number of letters counted with multiplicity.
\end{corollary}

\section{Questions and future directions}
There are a number of related questions that remain open.  Is it possible to extend the work of Section 5, finding generating functions for words avoiding other cyclic patterns? Is there a formula generalizing Theorem \ref{vincgf} or Theorem \ref{cyclicvinc} to to count the number of occurrences of a given pattern of ones?  One might also look for a combinatorial interpretation of some form of {\it composition} of Laguerre series; empirically, it seems that $l_i(l_j(t))$ has nonnegative integer coefficients in the $l_k$-basis for $j>0$. Finally, it would be useful to develop bijections from sets of words with restrictions to other combinatorial objects that are not obviously described in terms of words, using the methods outlined here to count sets that may be otherwise difficult to enumerate.



\bibliographystyle{plain}	% (uses file "plain.bst")




%\bibliography{myrefs}		% expects file "myrefs.bib"

\end{document}
